Author's response to reviews

Title: Effect of Physical Activity and Sun Exposure on Vitamin D Status of Saudi Children and Adolescents

Authors:

Abdulaziz Al-Othman (a_alothman@hotmail.com)
Sara Al-Musharaf (sara.almosharruf@gmail.com)
Nasser Al-Daghri (aldaghri2011@gmail.com)
Soundararajan Krishnaswamy (ksounder@gmail.com)
Deqa Yusuf (markeezaso12@live.co.uk)
Khalid Alkharfy (alkharfu@ksu.edu.sa)
Yousef Al-Saleh (alaslawi@hotmail.com)
Omar Al-Attas (omrattas@ksu.edu.sa)
Majed Alokail (msa85@yahoo.co.uk)
Osama Moharram (biomarkers@ksu.edu.sa)
Shaun Sabico (eaglescout01@yahoo.com)
George Chrousos (chrousos@gmail.com)

Version: 2 Date: 21 February 2012

Author's response to reviews:

To the Editor of BMC Pediatrics,

This manuscript was previously submitted to Cardiovascular Diabetology.